**Author’s response to reviews**

**Title:** Renal Association Clinical Practice Guideline on Monitoring children and young people with, or at risk of developing Autosomal Dominant Polycystic Kidney Disease (ADPKD)

**Authors:**

Melanie Dillon (melanie.dillon@renalregistry.nhs.uk)

Jan Dudley (jan.dudley@nhs.net)

Paul Winyard (p.winyard@ucl.ac.uk)

Matko Marlais (m.marlais@ucl.ac.uk)

Oliver Cuthell (ollycuthell@yahoo.co.uk)

Tess Harris (tess.harris@pkdcharity.org.uk)

Jiehan Chong (j.chong@sheffield.ac.uk)

John Sayer (john.sayer@ncl.ac.uk)

Daniel Gale (d.gale@ucl.ac.uk)

Lucy Moore (Lucy.moore3@btinternet.com)

Kay Turner (kay.turner13@yahoo.co.uk)

Sarah Burrows (Sarah.Borrows@uhb.nhs.uk)

Richard Sandford (rns13@medschl.cam.ac.uk)

**Version:** 1  **Date:** 27 Dec 2018

**Author’s response to reviews:**

The authors have requested that only my email is shown as a lot of spam messages come through as a result of these submissions and it's best if they all just come to me. I included all the author's emails in the online submission.

I've re-uploaded the manuscript with combined references.

I've removed the Table of Contents.

The lay summary can be our abstract.